Author's response to reviews

Title: The effects of multi-domain versus single-domain cognitive training in non-demented older people: a randomized controlled trial

Authors:

Yan Cheng (yancheng_research@hotmail.com)
Wenyuan Wu (wenyuanwu_tj@hotmail.com)
Wei Feng (ffww06@163.com)
Jiaqi Wang (jiamian316@163.com)
You Chen (huaganyan@163.com)
Yuan Shen (shenyuanmm@yahoo.com.cn)
Qingwei Li (lianocd@yahoo.com.cn)
Xu Zhang (zzz1082@163.com)
Chunbo Li (chunbo_li@yahoo.com)

Version: 4 Date: 18 February 2012

Author's response to reviews: see over
Dear Dr. Tree-Booker,

Thank you for the comments (MS: 5651524986181287). We have modified the manuscript according to your suggestions. All the changes we have made are typed in red.

Sincerely yours,

Wenyuan Wu

Question 1:

RESULTS: Recruitment Dates defining the periods of recruitment and follow-up
Why the trial ended or was stopped
- You have included the intended dates of recruitment in the Methods section but they must also be included in the Results section.

Reply: We really appreciated your suggestions. We added the related content in the Demographic information section in Page 13. Thank you very much.

Question 2:

DISCUSSION: Limitations. Trial limitations, addressing sources of potential bias, imprecision, and, if relevant, multiplicity of analyses
- You have discussed the limitations, but need to address the sources of bias.

Reply: Thank you for the suggestion. We have checked and revised the manuscript according to the guidelines in Page 21. Thank you very much.

Question 3:
Please make sure that all figure legends contain all the abbreviations used in the figures (e.g. RBANS etc).

**Reply:** We are very grateful for your comments. We have added the abbreviations as suggested. Thank you very much.